Author's response to reviews

Title: Estimation of glomerular filtration rate by a radial basis function neural network in patients with type-2 diabetes mellitus

Authors:

Xun Liu (Lihuihui1970@163.com)
Linshan LI (Wangjiaheng1970@163.com)
Jinxia CHEN (Tangbeibei1970@163.com)
Mu-hua CHENG (Cuiyifan2006@163.com)
Linseng LV (Tangfeifei1111@163.com)
Cheng WANG (Fangxiaoheng2012@163.com)
Cheng-gang SHI (Heweiyi1970@126.com)
Hui PENG (Zhanga1970@163.com)
Cai-lian CHENG (Jiadd2012@163.com)
Hua TANG (Vin0710@126.com)
Zhujiang CHEN (VC0710@126.com)
Ming LI (Vivi0710@126.com)
Zengchun YE (Baire0710@126.com)
Jun ZHANG (Air0710@126.com)
Chanming LI (Pair0710@126.com)
Xiaoming WU (Vair0710@126.com)
Tan-qi Lou (lxun111@126.com)

Version: 2 Date: 17 December 2012

Author's response to reviews: see over
Dear Editor,

Greetings!
We changed the email address of the corresponding author because the previous one is out of use. This information was also updated in the title page of the manuscript.

Sorry for the inconvenience.
And thanks for your help!

Sincerely,

Tanqi Lou